Author’s response to reviews

Title: Companion dog acquisition and mental well-being. A community-based three-arm controlled study

Authors:
Lauren Powell (lauren.powell@sydney.edu.au)
Kate Edwards (kate.edwards@sydney.edu.au)
Paul McGreevy (paul.mcgreevy@sydney.edu.au)
Adrian Bauman (adrian.bauman@sydney.edu.au)
Anthony Podberscek (anthony.podberscek@sydney.edu.au)
Brendon Neilly (neillybrendon19@gmail.com)
Catherine Sherrington (cathie.sherrington@sydney.edu.au)
Emmanuel Stamatakis (emmanuel.stamatakis@sydney.edu.au)

Version: 3 Date: 08 Oct 2019

Author’s response to reviews:

Editor Comments:
We notice that author BN is missing from the authors' contributions section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

https://protect-au.mimecast.com/s/C_GNC0YZWVFvA684Sw4vue?domain=biomedcentral.com

Thank you. We have now updated the Authors’ Contributions section.

“ES initiated and designed the study, acquired funding, led the broader research program, and supervised the PAWS pilot study. KE, PM, AB, CS, and AP contributed to the study design. LP led the data collection. BN contributed to the acquisition of data. LP cleaned and analysed the data, drafted the manuscript, and carried out multiple rounds of revisions. ES redrafted parts of the manuscript. All authors read and approved the final version of the manuscript before submission.”

--Please remove the company names (e.g. Petrescue and Gumtree) from the methods section.
We have now removed the company names from the methods section.

Reviewer reports:

Maryam Akbarilakeh (Reviewer 1): Dear authors, Thank you. I read highlighted and the revised version of manuscript.

Regards

Thank you.